Author’s response to reviews

Title: Effectiveness of endometrial resection transcervical hysteroscopic based preventive treatment on the recurrence of endometrial polyps in post-menopausal women.

Authors:

Jesus S Jimenez-Lopez (jjimenez.hdsc@salud.madrid.org)
Ana Granados-Sanmiguel (angranado@gmail.com)
Alvaro Tejerizo-Garcia (tejerizo@hotmail.com)
Jose L Muñoz-Gonzalez (jlmunozg@yahoo.es)
Gregorio Lopez-Gonzalez (goylopez2@hotmail.co)

Version: 5 Date: 11 October 2014

Author’s response to reviews: see over
Dear Prof O’Donovani:

We attach article entitled:
"Effectiveness of hysteroscopic endometrial transcervical resection preventive treatment based on the recurrence of endometrial polyps in post-menopausal women."

Similarly we have followed the recommendations stated in his email us the Monday, September 22, 2014.

We have included the Ethics statement:
"This protocol was submitted and approved by the committee of our clinical department and the ethical committee of our hospital, providing both agencies approval (Ethics Committee of the Hospital 12 Octubre), and complies with the Declaration of Helsinki. After the procedure and follow-up time periods were explained, participation was entirely voluntary, all patients signed informed consent forms."

We have reviewed and adjusted the text to publication standards. And finally we have defined the Authors' Contributions as:

"Authors’ contributions
JSJL and AGS designed the study, analyzed the data, and wrote the manuscript. GLG, JLMG and ATG contributed to study design, coordination, data analysis, and manuscript writing. JSJL coordinated the study, formatted data, and contributed to data analysis and manuscript editing. All authors read and approved the final manuscript"

Thank you in advance for your attention to our contribution.

Yours sincerely,

Jesús S. Jiménez-López, MD, PhD
Service of Obstetrics and Gynecology. Hospital Universitario 12 de Octubre
Avenida de Córdoba s/n. E- 28041 Madrid, Spain. Tel. and fax: +34 91 3908376
E-mail: jjimenez.hdoc@salud.madrid.org, jjimenezl@salud.madrid.org and jjimenezme35426@gmail.com